Author’s response to reviews

Title: A novel SMARCAL1 mutation associated with a mild phenotype of Schimke Immuno-Osseus Dysplasia (SIOD)

Authors:

Luisa Santangelo (luisa.santangelo@hotmail.it)
Maddalena Gigante (maddalena.gigante@unifg.it)
Giuseppe S Netti (giuseppestefano.netti@unifg.it)
Sterpeta Diella (esterbarletta@hotmail.it)
Flora Puteo (flora.puteo@policlinico.ba.it)
Vincenza Carbone (vincenza.carbone@policlinico.ba.it)
Giuseppe Grandaliano (giuseppe.grandaliano@unifg.it)
Mario Giordano (mariogiordanobari@fastwebnet.it)
Loreto Gesualdo (loretoge60@gmail.com)

Version: 2 Date: 17 September 2013

Author’s response to reviews:

Dear Editor,

enclosed please find the revised version of manuscript entitled “A novel SMARCAL1 mutation associated with a mild phenotype of Schimke Immuno-Osseus Dysplasia (SIOD)” by Luisa Santangelo, Maddalena Gigante, Giuseppe Stefano Netti, Sterpeta Diella, Flora Puteo, Vincenza Carbone, Giuseppe Grandaliano, Mario Giordano and Loreto Gesualdo, for submission as Case Report to BMC Nephrology.

GenBank accession codes of SMARCAL1 gene sequence has been added in “case presentation section”.

We hope that the enclosed manuscript will be now suitable for peer review process Best Regards,

Mario Giordano, MD

Correspondence should be addressed to:

Mario Giordano, MD
Pediatric Nephrology and Dialysis Unit
Ospedale Pediatrico Giovanni XXIII
Viale Giovanni Amendola, 207 - 70126 Bari (Italy)
E-mail address: mariogiordanobari@fastwebnet.it